Author's response to reviews

Title: Intrathoracic fire during preparation of the left internal thoracic artery for coronary artery bypass grafting

Authors:

Martin Friedrich (m.friedrich@med.uni-goettingen.de)
Theodor Tirilomis (theodor.tirilomis@med.uni-goettingen.de)
Jan D Schmitto (schmitto@med.uni-goettingen.de)
Aron F Popov (popov@med.uni-goettingen.de)
Suyog A Mokashi (mokashi@yahoo.com)
Marc Hinterthaner (hinterthaner@med.uni-goettingen.de)
Gunnar G Hanekop (gunnar@hanekop.de)
Paul Zwaka (paul_zwaka@yahoo.de)
Friedrich A Schoendube (schoendube@med.uni-goettingen.de)

Version: 4 Date: 5 March 2010

Author's response to reviews:

Dear Editor:

Author list changes are done - Authors’ contributions are inserted - table now without shading and colours - reference 15: excuse for the forgotten page numbers: they are now inserted - Title sentence case is uded - We found no typographical errors

Best wishes an Thank You - Martin Friedrich